Author's response to reviews

Title: Colon Capsule Endoscopy compared to Conventional Colonoscopy under routine screening conditions

Authors:

Julia B Pilz (julia.pilz@ksli.ch)
Susanne Portmann (susanne.portmann@stud.unibas.ch)
Shajan Peter (shajan@uab.edu)
Christoph Beglinger (Beglinger@tmr.ch)
Lukas Degen (ldegen@uhbs.ch)

Version: 2 Date: 10 December 2009

Author's response to reviews: see over
None of the authors has served as a speaker, a consultant and an advisory board member for Given Imaging, Yoqneam, Israel, none is an employee of, owns stocks and/or shares in or owns a patent from Given Imaging, Yoqneam, Israel.

Declaration of funding interests: The study was partly funded by the Nycomed Fund of the University Hospital Basel, Switzerland, and by Given Imaging Ltd., Yoqneam, Israel.

Acknowledgements: The authors thank Prof. Dr. Jürgen Drewe, University Hospital Basel, for Review of Statistical Analysis. J.B.P. and S.P. contributed equally to this study.

No possible conflict to the exclusive nature of this publication. Preliminary results have been published as:

- An oral session at the 7th International Conference on Capsule Endoscopy (ICCE) and 1st International Workshop on Colon Capsule Endoscopy Berlin, Germany, June 6-7th, 2008
- A poster at the Annual Assembly of the Swiss Gastroenterological Association (SGG-SSG), Interlaken, Switzerland, September 28-29th, 2008
- A poster at the 18th United European Gastroenterological Week (UEGW) Vienna, Austria, October 18-22th, 2008